Author's response to reviews

Title: MSH6 and PMS2 mutation positive Australian Lynch syndrome families: novel mutations, cancer risk and age of diagnosis of colorectal cancer

Authors:

Bente A Talseth-Palmer (Bente.Talseth-Palmer@newcastle.edu.au)
Mary McPhillips (Mary.McPhillips@hnehealth.nsw.gov.au)
Claire Groombridge (Claire.Groombridge@hnehealth.nsw.gov.au)
Allan Spigelman (aspigelman@stvincents.com.au)
Rodney J Scott (Rodney.Scott@newcastle.edu.au)

Version: 5 Date: 18 May 2010

Author's response to reviews:

Dear HCCP editor,

Re: Final confirmation

One sentence in the abstract has been reformulated:
From -"The risk of developing CRC is different between males and females which is 34% at age 50 years for males, while for females it is 21%.”
To - The risk of developing CRC is different between males and females at age 50 years, which is 34% for males and 21% for females.”

Kind regards,

Bente Talseth-Palmer